Author’s response to reviews

Title: To strengthen self-confidence as a step in improving prehospital youth laymen basic life support

Authors:

Anna Abelsson (anna.abelsson@ju.se)
Per Odestrand (per.odestrand@ju.se)
Annette Nygårđh (annette.nygardh@ju.se)

Version: 4 Date: 26 Dec 2019

Author’s response to reviews:

Thank you for your suggestions to our manuscript. We are grateful for the constructive and encouraging comments from the reviewers. We hope that you will find the revised manuscript improved and suitable for publication.

REQUESTED REVISIONS:

Still I have 2 suggestions:

1) The conclusions need to follow from the data. Beyond the first sentence of the abstract conclusion, the rest is just speculative.

The conclusion is based on the result integrated with the discussion. Facts that emerge lead us to draw conclusions on how self-confidence should be improved in young people

2) The discussion is still too long and should be shortened. No need to repeat the results in the discussion, just qualitative description and description of implications in the context of the literature. We have tried to shorten the discussion without losing valuable discussion.